Author's response to reviews

Title: Foot health education for people with rheumatoid arthritis: the practitioner's perspective.

Authors:

Andrea S Graham Mrs (a.s.graham@salford.ac.uk)
Alison Hammond Mrs (a.hammond@salford.ac.uk)
Anita E Williams Mrs (a.e.williams1@salford.ac.uk)

Version: 6 Date: 4 January 2012

Author's response to reviews: see over
Dear Independent Editorial Production Team,

Thank you to the editorial team for your comments and advised amendments. Please find below a point-by-point description of the changes made to the above manuscript in response to the formatting changes required, in the order that they were originally listed:

Amendments requested:

Major revisions (we require the author to make these changes)

Abstract: This should be identical in the manuscript and the submission system. Currently some of the tracked changes appear in the submission system abstract that will not appear in the final version of the manuscript. – The abstract in the submission system is identical to that of the manuscript and without tracked changes.

References: Please provide a functioning link to reference 17. If this is not possible then please indicate the date the site was last accessed. – This reference has now been removed from the list and the reference numbers in the list and manuscript have been amended accordingly.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Qualifications: Please remove all qualifications/titles from the author list on the submission system. – There are no qualifications/titles in the author list on the submission system.
Highlighting/tracking: Please remove all highlighting and tracking from the manuscript. Please ensure all text is black. — all highlighting, tracking has been removed and all the text is black.

Figures: The image file should not include the title (e.g. Figure 1... etc.) or figure number. The legend and title should be part of the manuscript file after the reference list. The figures are numbered automatically in the order in which they are uploaded. — The figure has been re-uploaded so that the title does not include the word ‘figure’, the Figure has been listed in the manuscript after the reference list according to the guidance provided and has been formatted so that there is minimal white space around the figure.

I hope this is a detailed enough description of the changes made. If you require any additional information, I would be more than happy to provide it.

Yours sincerely,

Andrea Graham, Lecturer in Podiatry.